Author's response to reviews

Title: Patient understanding of moles and skin cancer, and factors influencing presentation in primary care: a qualitative study

Authors:

Fiona M Walter (fmw22@medschl.cam.ac.uk)
Elka Humphrys (eh358@medschl.cam.ac.uk)
Simon Tso (simontso@doctors.org.uk)
Margaret Johnson (ottersriverbank@aol.com)
Simon Cohn (simon.cohn@medschl.cam.ac.uk)

Version: 7 Date: 27 August 2010

Author's response to reviews:

Only one change has been made: in the results section of the abstract the number 4 has been replaced with the word 'four'.